

Strategies to advance the dream of integrated digital public service delivery in interorganizational collaboration networks

Wouters, Stijn; Janssen, Marijn; Lember, Veiko; Crompvoets, Joep

10.1016/j.giq.2022.101779

Publication date

Document Version Final published version

Published in

Government Information Quarterly

Citation (APA)

Wouters, S., Janssen, M., Lember, V., & Crompvoets, J. (2022). Strategies to advance the dream of integrated digital public service delivery in inter-organizational collaboration networks. *Government Information Quarterly*, *40*(1), Article 101779. https://doi.org/10.1016/j.giq.2022.101779

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright

Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

ELSEVIER

Contents lists available at ScienceDirect

Government Information Quarterly

journal homepage: www.elsevier.com/locate/govinf





Strategies to advance the dream of integrated digital public service delivery in inter-organizational collaboration networks

Stijn Wouters ^{a,*}, Marijn Janssen ^b, Veiko Lember ^{a,c}, Joep Crompvoets ^a

- a KU Leuven, Leuven, Belgium
- ^b Delft University of Technology, Delft, the Netherlands
- ^c Tallinn University of Technology, Tallinn, Estonia

ARTICLE INFO

Keywords: Integrated service delivery Collaboration strategies Inter-organizational collaboration E-government Adoption paradox

ABSTRACT

Public administrations are increasingly relying on collaboration within inter-organizational networks to coordinate the development and provisioning of integrated digital public services. Collaboration strategies are necessary to focus collaborative efforts, align and prioritize goals, and leverage concrete results that advance integrated service delivery (ISD). However, literature on inter-organizational collaboration strategies for integrated service delivery is scarce. This research identifies collaboration strategies in three qualitative case studies in Belgium. The cases present 33 collaboration strategies categorized into engagement, motivation, and joint action strategies. Collaboration strategies should complement each other. Together these strategies can help to overcome the adoption paradox, one of the key issues to advance ISD, which refers to service providers not being inclined to adopt an ISD unless it has many users, but users might not be willing to use and ISD before many service providers have adopted it. Policy-makers can use the identified collaboration strategies to advance integrated service delivery.

1. Introduction

Integrated (digital public) service delivery (ISD) is a long-held dream, but a hard to realize goal of public administrations and remains a central research focus (Boudreau & Bernier, 2017; Wimmer, 2002). ISD relies on the integration of previously separated public services into public service chains through the use of common interfaces and shared building blocks (e.g., identity and access management, common portals) provided by different parties (Bharosa, van Wijk, de Winne, & Janssen, 2015; Van Veenstra, Janssen, & Klievink, 2009). Examples of ISDs include combined permits or life events that can be delivered through one-stop-shops (Wimmer, 2002).

In a public administration context where service delivery is expanding beyond organizational, sectoral and territorial boundaries, ISD can be viewed as a complex web of loosely coupled building blocks provided by various inter-organizational networks (Chen, 2008). Collaboration between public legal entities in inter-organizational networks has become the de facto coordination approach to advance digital public service provision (Chen, Hu, Tseng, Juang, & Chang, 2019; Fan, Liu, Huang, & Zhu, 2019; Provan & Milward, 2001). In this context,

collaboration refers to "activities in which two or more agents work together to achieve shared goals" (Chun, Luna-Reyes, & Sandoval-Almazán, 2012, p. 6). The benefits of inter-organizational collaboration regarding service integration have been widely acknowledged in the literature and in particular for e-government (e.g., Agranoff, 1991; Dawes, Cresswell, & Pardo, 2009; Einbinder, Robertson, Garcia, Vuckovic, & Patti, 2000). Benefits relate to improved service delivery, better goal attainment, and the sharing of resources (Agranoff, 1991; Dedrick & Greenbaum, 2011; Einbinder et al., 2000).

Chen et al. (2019) posit that inter-organizational collaboration and service integration is also driven by a more comprehensive view of public service delivery. Such a joined-up service delivery prioritizes a public service provision in which users are no longer responsible for linking different public services together. ISDs follow this evolution. Regarding ISD, inter-organizational collaboration is specifically needed to establish the technical infrastructure, to integrate back-office systems and to stimulate adoption by users (Juell-Skielse, Lönn, & Päivärinta, 2017). Collaboration requires building trust so that public service providers are willing to work together to overcome differing expectations (Luna-Reyes, Gil-Garcia, & Cruz, 2007; Vangen & Huxham, 2003). At

^{*} Corresponding author at: Parkstraat 45, 3000, Leuven, Belgium.

E-mail addresses: Stijn.wouters@kuleuven.be (S. Wouters), M.F.W.H.A.Janssen@tudelft.nl (M. Janssen), Veiko.lember@taltech.ee (V. Lember), Joep. crompvoets@kuleuven.be (J. Crompvoets).

the same time, advancing ISD through inter-organizational collaboration is also hard to achieve due to the diversity of parties, diverse requirements, and the need for the development of infrastructures (Wouters, Janssen, & Crompvoets, 2020). ISD creates dependencies, including shared resources, making it harder for public legal entities to attain managerial and political objectives. These objectives might differ from the objectives of the ISD, and make public legal entities reluctant to give up their autonomy (Boudreau & Bernier, 2017; Van Veenstra et al., 2009; Yildiz, 2007).

In the last decades, a significant body of research has emerged regarding e-government collaboration. This body of research has mainly considered systems of information exchange and information sharing (e. g., Gil-Garcia, Guler, Pardo, & Burke, 2019; Pardo, Gil-Garcia, & Luna-Reyes, 2010) and systems of inter-organizational information integration (e.g., Chen et al., 2019; Luna-Reyes et al., 2007). Collaborative efforts concerning ISD are still an under-researched topic (Boudreau & Bernier, 2017; Lang & Brüesch, 2020). The e-government collaboration literature has also predominantly focused on examining factors that stimulate collaboration (e.g., trust, shared goals, clear goals and responsibilities) or that are detrimental to it (e.g., negative prior experience or a lack of funding) (e.g., Chen et al., 2019; Gil-Garcia et al., 2019). What has been lacking is an examination of collaboration strategies to spur, utilize or mitigate digital government (see e.g., Van Veenstra et al., 2009). A strategy concerns setting long-term goals and defining actions to achieve those goals (Johnson, Scholes, & Whittington, 2008). The long-term goal is a collaboration among parties needed for creating ISD. Collaboration strategies consist of concrete actions that aim to focus collaborative efforts or lead to concrete joint actions that advance ISD. While some specific strategies, such as providing direct and indirect incentives for participation, have been identified in the egovernment collaboration literature (Pardo et al., 2010; van Os, 2011), a comprehensive overview of collaboration strategies is still lacking.

This research contributes to the e-government literature by providing a comprehensive empirical insight into existing collaboration strategies in ISD. Hence, the research question in this paper is: What are inter-organizational collaboration strategies to advance ISD?

A multiple-case study approach is adopted to identify collaboration strategies. We draw on findings from three ISD cases in the region of Flanders, Belgium. Together they cover a wide range of collaboration strategies. The remainder of this paper is structured as follows. Section 2 contains the research background on collaboration strategies and ISD. Section 3 details the research method, while section 4 provides information on the cases. The results are presented in section 5 and discussed in section 6. Section 7 provides the conclusion.

2. Research background

2.1. Integrated service delivery

Although ISD has been the subject for research since the beginning of e-government research (e.g., Wimmer, 2002) and even before digital government initiatives (e.g., Agranoff, 1991; Waldfogel, 1997), there is less consensus about what it entails. We consider ISD as a subset of digital public service delivery. Following Lindgren and Jansson (Lindgren & Jansson, 2013, p. 167), we view a digital (or electronic) public service as a service in the public sphere that primarily relies on electronic mediation between actors. Digital public services undergo different development phases, such as initiation, piloting, operationalization and expansion (Wouters, Janssen, & Crompvoets, 2020). While the strategic objectives or core functionalities largely remain the same, digital public services can change throughout these phases. For example, when incorporating new technologies, implementing new regulations, or adding functionalities.

The key feature that distinguishes ISDs from other types of digital public services is integration. An ISD consists of a group of public services from different public service providers linked together in a coherent manner in service delivery chains (Bharosa et al., 2015). Integration thus follows from the capability of public services (especially their interfaces, such as websites) to provide a single service, irrespective of the all the public service providers that delivering (parts of) the service (Puron-Cid et al., 2022). An ISD can also include private service providers. For example, private legal entities acting as intermediaries (Janssen & Klievink, 2009). These service delivery chains are orchestrated to provide ISD users with an integrated view and a seamless interaction (Klievink & Janssen, 2008; Van Veenstra et al., 2009). ISDs reduce the number of interactions and the need for users to negotiate with public service providers directly and separately, thus lowering the administrative burden (Van Veenstra & Janssen, 2011).

2.2. Inter-organizational collaboration and collaboration strategies

ISDs require organizations to collaborate together to integrate separate public services (Boudreau & Bernier, 2017). Collaboration in this paper concerns organizations working together to attain common goals (Kettl, 2006). In this research, collaboration refers to the process of organizations working together for the goal of establishing and modifying ISDs. By collaborating, organizations adjust their own internal business processes and systems to create a public service that is perceived by the user as an integrated whole (an ISD). The literature has paid attention to various forms of collaboration, including cross-sector collaborations (Bryson, Crosby, & Stone, 2006; Emerson & Nabatchi, 2015; Klievink, Bharosa, & Tan, 2016) and inter-organizational collaboration (Agranoff, 2017; Fan et al., 2019; Juell-Skielse et al., 2017; Luna-Reyes et al., 2007).

This paper takes the e-government literature that has developed around inter-organizational and inter-agency collaboration as a starting point to explore inter-organizational collaboration in ISD (e.g., Chen et al., 2019; Chun et al., 2012; Gasco-Hernandez, Gil-Garcia, & Luna-Reyes, 2022; Picazo-Vela, Gutiérrez-Martínez, Duhamel, Luna, & Luna-Reyes, 2018). Collaboration allows the involved organizations "to solve differences, identify interdependencies and build ownership of decisions" (Chun et al., 2012, p. 6). Through collaboration, organizations can exchange resources, such as knowledge or technology (Picazo-Vela et al., 2018), but also work out shared problems, such as ISD (Chen, 2008).

Inter-organizational collaboration is not easy to accomplish (Gil-Garcia, 2012; Provan & Milward, 2001; Scholl & Klischewski, 2007), whether in a public administration context in general, or in a specific digital government setting that often includes long-term engagements through the integration of information systems (Juell-Skielse et al., 2017; Scholl & Klischewski, 2007). Differing expectations and preferences regarding public values, insufficient expertise and capabilities, a lack of consensus concerning the division of roles and responsibilities and the absence of trust between collaborators (inter alia) can pose considerable challenges (Chen & Lee, 2018; Juell-Skielse et al., 2017).

Collaboration necessitates organizations to balance autonomy, identity and cohesion (Chen, 2008). Boudreau and Bernier (2017) indicate how collaborative efforts need to overcome public service providers' institutionalized attitudes towards autonomy and find ways to integrate (parts of) their public services. As Lundin (2007) found, joint action will only be fruitful by addressing mutual trust and goal congruence simultaneously. Emerson and Nabatchi (2015) argue that collaboration has to focus on three elements: (1) finding shared goals and agreeing on the methods for deliberation and determining actions (engagement), (2) motivate actors to start or to continue collaborating (shared motivation), and (3) providing capacities such as knowledge and resources that allow actions to be taken (capacity for joint action).

Collaboration strategies are necessary to approach these elements, bring focus to collaborative efforts, prioritize activities, and leverage concrete results. Strategies consist of long-term goals on the one hand and defining actions on the other hand (Johnson et al., 2008). In this article, objectives are stated in relation to advancing ISD. Those

objectives can be expressed in broad strategy statements such as establishing and gradually expanding the digital infrastructure of an ISD (i.e., adding functionalities), and stimulating adoption by users and public service providers. Collaboration can be considered a specific strategic action to improve policy outcomes in inter-organizational networks (Bryson & George, 2020; Imperial, 2005). Collaboration strategies in this research are actions that rely on collaboration. Those actions are directed at advancing the objectives of ISD specified above (i.e., the outcome of the collaboration) and/or at organizing the collaboration and encouraging the motivation of public service providers to participate in the collaborative efforts (i.e., the process of the collaboration). Strategies can be both deliberate and emergent, the latter meaning that strategies can be unplanned and can even be unrecognized (Bryson & George, 2020). In this paper, we look at both types of strategies in order to go beyond deliberate strategies noted by lead organizations and get a comprehensive overview of collaboration strategies for ISD.

2.3. Collaboration strategies for ISD

In the e-government literature, examples of collaboration strategies for ISD can be found, but they are often a side finding and often mentioned as part of the discussion. This section reviews the literature on e-government collaboration to gain insights into collaboration strategies. Studies regarding e-government collaboration and ISD are examined in particular.

Van Veenstra et al. (2009) identified four architecture design strategies to achieve ISD related to the architecture of services and the level of control by the parties involved: merger, orchestra, relacy race, and broadcasting. For each of these architecture design strategies the authors describe the autonomy of the various parties involved. These strategies might be used to spur collaboration in several ways. They can, for example, be applied to overcome fears of losing public service providers' autonomy or to get around the organizational unreadiness of potential public service providers to integrate their services.

For public–private information platforms, Klievink et al. (2016) found collaboration strategies including deliberation to come to voluntary adoption of standards, collective business model designs to promote adoption of the service, and awareness creation related to potential advantages to the different parties involved.

Gil-Garcia, Chengalur-Smith, and Duchessi (2007) propose a collaboration strategy based on goal-orientation rather than micromanagement, with clear and realistic goals. In order to deal with the expansion of the involved public service providers, the authors favor a more incremental approach with rapid (partial) results to demonstrate the collaboration's viability (Gil-Garcia, Schneider, Pardo, & Cresswell, 2005). Discussing the potential that different technologies offer can be a collaboration strategy to establish a common vision and set ambitious goals (Chen & Lee, 2018). van Os (2011) advances a business case as the central object for public legal entities that can be used to establish a consensus on the objective and scope of collaboration among collaborating public legal entities. Mutual adjustment through interaction and communication is another potential strategy used to form a consensus and find a common motivation to integrate existing ISDs (van Os, 2011).

Establishing clear roles and responsibilities can be a strategy intended to reduce resistance among participants in collaborative efforts (Gil-Garcia et al., 2019). Building on the work of Sayogo, Gil-Garcia, and Cronemberger (2016), collaboration strategies to pursue clear roles and responsibilities can consist of frequent interactions; being considerate of the diversity of participants' contexts, objectives and goals; and the use of boundary objects (Lönn & Uppström, 2016), with formal rules and procedures set at the initiation to guide collaboration (Gil-Garcia et al., 2019). As Gil-Garcia et al. (2019) note, the downside of more formalized relationships may come at the cost of having adequate flexibility to deal with new issues. Klievink et al. (2016) point to a division of roles that includes lead organizations. Those lead organizations can develop interface standards and provide technical guidance. Similarly, Gil-

Garcia and Sayogo (2016) advocate for formally assigned project managers.

The literature has well-established the need for strategies that induce trust among collaborating public legal entities (Chen, 2008; Chen et al., 2019). Clear roles and responsibilities have also been favored as a strategy for trust (e.g., Gil-Garcia et al., 2019). Trust can also be achieved by adequate communication (Chen & Lee, 2018), or by making clear the expected benefits, showing quick results, and using institutional arrangements such as laws and regulations (Luna-Reyes et al., 2007).

Other collaboration strategies relate to the use of incentives. van Os (2011) details direct incentives such as decreasing or increasing budgets or indirect incentives such as benchmarking. According to van Os, indirect financial incentives can be a source to stimulate motivation. Chen and Lee (2018) add clear requirements and technical support to complement financial incentives to public service providers.

The sharing of information and knowledge has also been put forward as a strategy in the literature (Gil-Garcia & Sayogo, 2016; Pardo et al., 2010). Knowledge sharing, in combination with formal and informal relationships between public service providers can be collaboration strategies that contribute to joint decision-making capabilities (Janssen, Kamal, Weerakkody, & Joha, 2012). Yang and Maxwell (2011) propose legislation and regulation as possible instruments to enhance trust between actors to stimulate information sharing.

In summary, the background literature provides four elements and suggestions for collaboration strategies. First, the literature points to collaboration strategies related to a stepwise and incremental approach to awareness-raising, setting common goals, the use of incentives, clear roles and responsibilities, formal and informal relationships and procedures, and information and knowledge sharing. Second, the literature also shows that collaboration strategies can be used to further a variety of goals, such as increasing mutual adjustment to the objectives and scope of the ISD, joint decision-making, motivation and trust, and adoption. However, not all collaboration strategies that were found in the literature made this clear. Third, the review finds that a single collaboration strategy can have multiple goals. Fourth, collaboration strategies can be supported by the service architecture of an ISD or legislation. However, a comprehensive overview of collaboration strategies for ISD focusing both on deliberate and emergent strategic actions remains lacking.

3. Research method

The objective of this research is to identify collaboration strategies for ISD. We do this by identifying collaboration strategies in three cases. The methodology adopts an interpretative and pragmatic research philosophy (Goldkuhl, 2012). This means that we rely on the interpretations made by the involved professionals to apprehend interorganizational collaboration strategies (Walsham, 1995). The philosophy is pragmatic in the sense that the research is intended to identify collaboration strategies that have practical relevance.

A qualitative multiple-case study approach was adopted for this research (Yin, 2018). It is suited to answer a "what" question and look for patterns in the data (Bhattacherjee, 2012). A case study approach is justified because it allows us to scrutinize collaboration strategies in detail and consider the context in which they are applied (Swanborn, 2010). We adopt a multiple-case study to understand the complexity of ISD phenomenon and to find collaboration strategies beyond the results of a specific case. The unit of analysis is the coordination between public legal entities. The research approach is deductive-inductive (Miles & Huberman, 1994). Insights from the literature are taken as a starting point to study the observations in the case studies, which are then used to expand the current understanding of collaboration strategies.

The number of possible ISDs is large and diverse. We chose to examine cases of ISD that deliver services to external users (i.e., natural persons and private legal entities), follow the move towards a more

central service provisioning following reintegration efforts (Dunleavy, Margetts, Bastow, & Tinkler, 2005), and focus on cases that cross administrative boundaries because collaboration there is particularly challenging (e.g., Chen et al., 2019). Therefore, we applied the following selection criteria. The cases entailed ISD, aimed at (1) (at least) external users, (2) incorporated multiple levels of government, and (3) collaboration has to be a core component of the governance of the public services, rather than a (from the start) mandated adoption of an existing public service. The cases were selected based on known characteristics (Denscombe, 2014). Based on these criteria, we selected three cases: (1) Digital Invoicing, (2) eBox, and (3) My Citizen profile. In these cases, ISD aims to provide value to natural persons and private legal entities in the region of Flanders, the northernmost state in the federal country of Belgium. The cases take place in a single country. On the one hand, this limits cultural and historical variability (Juell-Skielse et al., 2017). On the other hand, it poses a limitation on generalizability. Each of the cases has reached its expansion phase. Initial objectives relating to functionalities have been achieved, new functionalities are being developed, and adoption of users and public services is high or increasing.

Two data collection techniques were used: a review of documents and semi-structured interviews. Public and internal documents allowed us to examine the context of the cases, as well as the formal collaboration structures and instruments. Documents included policy documents, policy memoranda and policy letters, business plans, governance agreements, legislation, regulations, parliamentary questions, websites and technical documentation (manuals, standards). Depending on the case, these documents were located at the European, federal, regional and/or local level. These documents guided the interview questions and later served as a source for triangulation. We conducted pilot interviews with the main collaborators for each of the three cases to get insights into the cases.

The data collection followed a cross-sectional time horizon. Data gathering took place between February and October 2020. A total of 52 interviews were done, respectively 14 for Digital Invoicing, 18 for eBox and 25 for My Citizen Profile. Five interviews were shared between the eBox Case and the My Citizen Profile Case at the request of the interviewees. Appendix A provides an overview of the interviews, the public legal entities and the functions of the interviewees. These interviews typically lasted between 60 and 120 min. The interviews were done through video-conferencing tools (due to COVID-19). Informed consent was obtained at the beginning of each interview. Interviews were recorded and transcribed. A purposive sampling strategy was followed to select the respondents (Ritchie, Lewis, McNaughton Nicholls, & Ormston, 2013). The criteria for purposive sampling of the respondents were (1) their affinity with the ISD case and (2) their involvement in the coordination structures. In order to be representative for the public legal entities involved in the collaboration and incorporating non-responders, a balance was sought with regards to federal, regional and local public legal entities, larger and smaller public service providers, and those involved at the initiation and later phases. Questions in the semistructured interviews related to the general context, actor roles, the setup of the public service and questions regarding the governance (e.g., strategic, legal and organizational aspects), as shown in Appendix B.

The analysis of the data is a thematic analysis (Braun & Clarke, 2006) and was carried out in NVivo. The goal of the analysis is to provide a rich description of the collaboration strategies in the data. The analysis entailed 6 steps: (1) familiarization, (2) initial coding, (3) searching for themes, (4) reviewing themes, (5) defining and naming themes, and (6) writing the analysis. Documentation and interviews were carefully read to get familiarized with the data. The analysis follows a process of deduction and induction. During the review of the documents and the pilot interviews, we found that collaboration strategies could be categorized into the three groups proposed by Emerson and Nabatchi (2015). For the initial coding, principal engagement, shared motivation, and joint action capacity were deductively used as sensitizing concepts to identify collaboration strategies. Because these concepts are still

abstract, the analysis was complemented with an inductive phase to search for additional themes within each sensitizing concept. Specific collaboration strategies were inducted from the data and grouped based on similarities. We carefully reviewed whether themes constituted new collaboration strategies or related to strategies as found in the literature. These three steps were done by the first researcher. The first and second researchers further refined, grouped, and/or renamed the collaboration strategies. The first researcher then wrote the analysis, in interaction with the other authors.

4. Case descriptions

This section describes the three cases that were selected to identify collaboration strategies for ISD in inter-organizational collaboration networks. Table 1 provides an overview of the characteristics of the three cases: Digital Invoicing, eBox and My Citizen Profile. These cases are embedded in the Belgian institutional and administrative context. Belgium is a federal country with (in contrast to other federations) two sets of federated states at the regional level: three communities (with competences such as education, culture and welfare) and three regions (responsible for e.g., housing, environment and mobility). Although Belgium is a federal country, it lacks typical federal characteristics such as a hierarchy of norms. A general intergovernmental e-government collaboration agreement and collaboration structure regarding digital government exist, but collaboration is mostly on a case-by-case. Each public administration at the federal and regional levels has one or more central public legal entities responsible for digital government. This paper focusses on three cases of ISD and the collaboration between and/ or the federal level, the regional level of Flanders (a single public administration responsible for both community and regional matters covering the northern part of the country) and the local level.

The Digital Invoicing Case involves the digitalization of invoices and related documents for public procurement purposes. The ISD was initiated in 2009. Several central federal building blocks have been developed to allow for a single entrance that exchanges invoices from private legal entities to public legal entities, and vice versa for outbound invoices. The entrance is linked to the PEPPOL network of private Access Points that aims to digitalize invoicing across sectors (see Van Donge, Bharosa, & Janssen, 2022). Public service providers connect to the platform through the regional or federal enterprise service bus (ESB). Multiple collaboration structures were created. These structures regard inter alia the exchange of invoices from private legal entities to public legal entities, the exchange between private legal entities, and at the regional level between Enterprise Resource Planners (ERPs). The standardization of invoices is linked to European Directive 2014/55/EU. Collaboration is mainly driven by two central public legal entities. These are the federal public legal entity responsible for digitalization (Federal Digital Transformation Office), and the regional public legal entity responsible for e-procurement (the Flemish ICT Support Agency). The federal public legal entity coordinates with the PEPPOL network and is as a public service intermediary responsible for the main building blocks specifically related to Digital Invoicing. A Reflection Working Group at the regional level is the main collaboration structure. It is led by the regional public legal entity and is composed of the regional public service intermediary that provides ERP services to most regional public legal entities, as well as and regional public service providers with separate ERPs. Collaboration at the federal level is on a bilateral basis. The federal Agency for Administrative Simplification has a role regarding adoption by private legal entities. Digital invoicing became gradually mandatory for public service providers and private legal entities (suppliers) after the ISD was developed, starting in 2017 (following and going beyond the European Directive).

The *eBox Case* has the goal of developing two-way communication for official correspondence between public legal entities, natural persons and private legal entities through a federated system of secure digital mailboxes. Initiatives related to digital mailboxes go back to 2005.

Table 1Case characteristics.

	Users	Public service inter	mediaries an	d public service providers	Main collaboration structure(s)
Digital Invoicing	Private / Public legal entities	Federal, regional, (local)	15+ involved	2 federal and 1 regional lead organizations	Regional collaboration structure including regional lead organization, regional public service intermediaries and public service providers
eBox	Natural persons Private / Public legal entities	Federal, regional, (local)	10+ involved	2 federal and 2 regional lead organizations	Federal collaboration structure including two federal lead organizations with public service intermediaries Regional collaboration structure between regional lead organizations and large regional public service providers Collaboration structures with 2 federal and 2 regional lead organizations
My Citizen Profile	Natural persons	Regional, local	$\begin{array}{c} 14 \rightarrow \\ 100 + \\ involved \end{array}$	1 regional lead organization	Regional collaboration structure with regional lead organization and regional public service providers

Collaboration regarding a federated system of digital mailboxes that provide an integrated view to users was initiated between 2014 and 2016. Natural persons can choose between a public eBox or private portals. The eBox for private and public legal entities integrates directly with public service providers (or indirectly through a private data service provider). For correspondence between public legal entities and private legal entities, one of the federal legal public entities for social security leads the collaboration (the Federal Social Security Agency). The regional entity for entrepreneurship (Flemish Agency for Innovation & Entrepreneurship) is a single point of contact for the regional and local legal public entities. The federal public legal entity for digitalization and its regional counterpart (Flemish Digital Agency) are the lead organizations for natural persons. The federal public legal entity is responsible for the eBox for natural persons. Besides roles regarding the portals, there are Document Providers (a public service intermediary that organizes databases for other public legal entities to store and make messages accessible) and Document Consumers (public service providers that send messages to a Document Provider). The two federal lead organizations and the regional digital agency also assume the role of Document Provider (among other public intermediaries). The eBox ISD has a federated collaboration structure. The two federal lead organizations organize a Document Provider forum. Each Document Provider coordinates with its own set of Document Consumers. Separate collaboration agreements exist between the regional and federal lead organizations that also encompass two separate collaboration structures. After the eBox ISD was developed at the beginning of 2019, operationalization has been underway to federal, regional and local Document Providers and (message generating processes of) Document Consumers.

The My Citizen Profile Case encompasses an ecosystem of building blocks that aims to provide natural persons a no-wrong-door access to public services of all local and regional public legal entities. The ISD was initiated in 2014 and became operational at the end of 2018. Its portal functionality can be integrated into regional and local communication channels (portals, websites, digital counters). The portal offers the starting point of informational and transactional digital public service provisioning. Other building blocks provide (1) a Single-Sign-On to other public services, (2) basic information regarding the user, and (3) several generic portal functionalities such as a feedback application concerning authoritative data, a notification application, a certificate application, and a file status application. The My Citizen Profile enables ISD for each of these functionalities, e.g., natural persons get all notifications from local and regional entities through the notification application. The regional public legal entity for digitalization is the lead organization of the program and is a public service intermediary between public service providers and users. The main collaboration structure is led by the regional lead organization and is composed of the regional and local public service providers. Since its operationalization, the processes of regional and local public service providers are gradually being integrated into the building blocks of the ISD.

5. Results¹

The cases reveal a wide range of collaboration strategies. Sections 5.1, 5.2 and 5.3 describe the collaboration strategies categorized into engagement, motivation and joint action collaboration strategies. Table 2 provides an overview of the 33 identified collaboration strategies. Appendix C details the frequencies of the mentioned collaboration strategies in the interviews. Some collaboration strategies were found in all cases, whereas others only in a single case. The latter suggests the need for further research to find all possible collaboration strategies. Section 5.4 shows how collaboration strategies across the three categories can be directed at specific issues. The section details this for the issue of the adoption paradox.

5.1. Engagement collaboration strategies

The first category of collaboration strategies is directed at fostering engagement between participants. Collaboration strategies in the Digital Invoicing Case were not much directed at engagement. The prospect of the legal (European) mandate and its implementation into federal and regional law made defining collaboration more straightforward. The initiation phase also concluded with clearly established mandates for the lead organizations and project structures based on executive decisions. The lead organizations organized an open deliberation process with public service providers. The two other cases did not have such strong legal or executive mandates, instead these cases relied on a business case to foster engagement among participants before the development of the ISD. The My Citizen Profile Case also built upon the broader project structure it was initially embedded in (the Flanders Radically Digital digitalization program). As part of the collaboration strategy 'Formal agreements at the management level to settle roles and responsibilities' the eBox Case and My Citizen Profile Case respectively use provider agreements and data sharing agreements to settle roles and responsibilities related to the processing of personal data.

A specific collaboration strategy found was related to the use of names. For the Digital Invoicing Case, at the regional level, the lead government organization convened a collaboration structure that was deliberately named 'Reflection working group' instead of 'Steering group'. This way, the lead government organization intended to downplay the importance of the collaboration structure to counter potential resistance of public service providers to participate.

"The goal of the group is actually to steer. Because, reflection

¹ This section makes use of quotes from the interviews. Quotes are attributed to individual interviewees by making use of codes. These correspond to interviews in Appendix A.

Table 2 Collaboration strategies.

Collaboration strates	gies	Digital Invoicing	eBox	My Citizen Profile
Related to	Open deliberation process of actors involved in collaboration efforts	Yes	Yes	Yes
engagement	Preserving the autonomy of public service providers through the architecture	No	Yes	Yes
	Business case to focus collaboration	No	Yes	Yes
	Formal agreements at the management level to settle roles and responsibilities	No	Yes	Yes
	Use cases, pilots and proof of concepts	Yes	Yes	Yes
	Deliberate use of names	Yes	No	Yes
	Demand oriented approach to deliberation, with priorities set by priorities by lead organization(s)	No	Yes	Yes
	Supply oriented approach to deliberation	Yes	No	Yes
Related to	Formal agreements at the management level to build trust and commitment	No	Yes	Yes
motivation	Finding common motivation and mutual understanding of each other's context	Yes	Yes	Yes
	Stimulate competition among early adopting public service providers	Yes	No	Yes
	Focusing on public services with large volumes or important public services	Yes	Yes	Yes
	Giving financial incentives	Yes	No	Yes
	Emphasizing lower burden for public service providers	Yes	Yes	Yes
	Use early adopters to demonstrate capability and build trust	Yes	Yes	Yes
	Being transparent about choices that are made	Yes	Yes	Yes
	Making clear long term continuity and finance	Yes	No	No
Related to joint	Dyadic collaboration by lead organizations	Yes	Yes	Yes
action	Separating business and IT collaboration	Yes	Yes	Yes
	Re-using already existing services	/a	In	Yes
			part	
	Change structural coordination approach across phases	No	Yes	Yes
	Flexible user-oriented architecture (regarding the users)	Yes	Yes	/
	Core functionalities and stepwise expansion	Yes	Yes	No
	Flexible architecture, approach and legacy management regarding (adoption by) public service providers	Yes	Yes	Yes
	Giving financial incentives to (early adopting) public service providers	Yes	No	Yes
	Provide clear roadmap(s) to focus and to anticipate changes	Yes	Yes	Yes
	Common support (for the integration of the various involved building blocks)	Yes	Yes	Yes
	Taking into account financial and technical capabilities	Yes	Yes	Yes
	Bilateral agreements (with public service providers / intermediaries / lead organizations)	Indirect	Yes	Yes
	Stimulating knowledge transfer between involved (or new) public service providers within and outside of	Yes	In	Yes
	collaboration		part	
	Providing varied documentation and technical support	Yes	Yes	Yes
	Using legal frameworks to come to institutional arrangements / settle legal uncertainty	No	Yes	Yes
	Ensuring a sustainable relation with IT-providers	In part	Yes	Yes

^a Not a viable collaboration strategy due to the architecture of the ISD.

working group, we gave that name to get it going, but actually, the working is that of a steering group. (D5.1) It does work, because that's what was reported. You see that there is a lot of ambition there. (D5.2) Yet, if we had called it a steering group from the onset, it wouldn't have come off the ground, I think. Because you immediately get someone who says: "Ow, what's going to happen there?" (D5.1).

In the My Citizen Profile Case, to come to a more supported definition of the goal and the scope, the name of the service was changed (before going into production) from 'My Citizen Counter' to 'My Citizen Profile' in the piloting phase. The reasoning was that "because a 'counter' meant too much of a takeover. It gave the impression of a takeover. And that is not what we wanted to express" (M3.1). Public service providers with more established 'brands' could feel threatened by the takeover and loss of a prominent digital communication channel. "It was a threat from the point of view of what already existed" (M4.1) one program managers explained.

5.2. Motivation collaboration strategies

The second category of collaboration strategies seeks to enhance mutual understanding, build trust, and preserve participants' commitment to collaboration.

In the eBox Case, formal collaboration agreements also served as a motivation collaboration strategy, meant to enhance trust between the different government levels with respect to the dependencies that would be created by the latter having to rely on the former to send official correspondence (such as messages related to taxes).

Impacting both the eBox and My Citizen Profile Cases, the COVID-19 crisis was instrumentalized as an ad hoc collaboration strategy for common motivation and spurred development and adoption among

intermediaries and public services providers to provide a rapid communication channel with natural persons. A lead organization noted that "we did see with the corona crisis and especially in the beginning, when the paper mail didn't work well, that there were many institutions that showed interest in getting on board." (E12.1).

Another collaboration strategy in the three cases was convincing public service providers of the benefits of ISD. While some public sector organizations were already motivated to contribute to ISD, others were not. Emphasis was placed at inter alia the benefits regarding a user-oriented approach related to ISD, external efficiency for natural persons and private legal entities as well as internal efficiency (i.e. emphasizing the total cost of ownership).

Giving financial incentives proved a viable collaboration strategy to stimulate commitment among (potential) public service providers. "The basic connection of Mercurius or PEPPOL, that cost is already fully borne by [the Federal Digital Transformation Office]. That's not being passed on, and in that area I think that's to us, everybody who makes use of it, is an incentive." (D10.1).

Unremarkable, this was related to stimulating the capacity of public service providers to integrate by ensuring adequate resources. In the Digital Invoicing Case, central financial resources were allocated to the lead regional public entity that financed the development and dedicated support services from the regional intermediary to the local and regional public service providers. "Digital invoicing is a bit of an exception, we [the Flemish Digital Agency] have a dedicated support contract with the Flemish ICT Support Agency." (D8.1) For the My Citizen Profile Case, financial incentives consisted of co-financing the initial integration of the portal functionality into the websites of public service providers. A local public service provider remarked that "co-financing, that's obviously a very easy one. We were lucky that we were indeed able to do that

integration with the header without it actually costing any money. So Flanders has looked at that smartly as far as I'm concerned." (M21.1).

5.3. Joint action collaboration strategies

The third category of collaboration strategies aims to develop the capacity for collaboration, and enhance the adoption of the ISD by public legal entities. Instead of a single intergovernmental collaboration structure the Digital Invoicing and the eBox Cases relied on several collaboration structures and bilateral approaches to come to joint action within and across inter-organizational networks. Due to the characteristics of Belgium's federal institutional system, there is almost no hierarchy between the federal level and the regional entities. Burdens on establishing formal inter-governmental cooperation agreements are also high. Furthermore, while ISD is stressed as important by each government, priorities between the federal administration and the regional entities differ.

The number of senders in the scope of the Digital Invoicing Case and the eBox Case includes all public legal entities in Belgium. In the Digital Invoicing Case bilateral collaboration between lead organizations is augmented with a collaboration structure bringing together the ERPs at the regional level. "You actually take care of a lot of complaints there. For example, there are problems with certain suppliers. [...] What is also important, you have some [public service providers] that are on different tracks. Other systems are added. Then they start helping each other. The federal government is always there, the people from [the Flemish Digital Agency] are there too. (D5.2).

The collaboration strategy in the eBox case is more complex. In addition to a joint action collaboration strategy focusing on bilateral lead organization collaboration, the case comprises federal collaboration structures regarding federal intermediaries, as well as collaboration structures between federal and regional lead organizations. While complex and especially time-consuming for the lead organizations, the federated collaboration approach is at the same time a collaboration strategy to keep the number of collaboration partners at the same time manageable. "We have a potential of - depending how you count - between 5000 and 6500 possible document senders. You can't possibly start having meetings with all those organizations." (D4.1) Service providers also saw this as beneficial as "you can stand with several against the other partner. [The Flemish Digital Agency] represents us. They almost represent the Flemish government, so to speak. If you are there as one entity, it is much more difficult. Even though we are an important player in certain matters." (M6.1) For both the Digital Invoicing and eBox cases, bilateral collaboration and a flexible architecture regarding public service providers allowed for the lead organizations to develop additional functionalities specific to or at the tempo of their own set of public service providers.

5.4. Collaboration strategies covering multiple categories

The cases reveal that collaboration strategies may have to be used simultaneously to address issues. Several collaboration strategies were used to mitigate the 'adoption paradox', a key issue that emerged from the cases. The adoption paradox concerns the relationship between the adoption of a public service by users and the adoption by public service providers. On the one hand, public service providers were hesitant to rely on ISD because adoption by the user remained low. On the other hand, lead organizations and public service providers worried that users would not be willing to adopt the digital service if not enough volume was created. Also, focusing on promoting the ISD without having a large enough volume of services available was thought to demotivate users to adopt the ISD. An interviewee put it in the following way: "You're only going to fill your box if there's also a lot of interest in it. Because if there's a lot of interest, then the pressure is higher to fill the box. That's the chicken or the egg." (M9.1).

In the eBox Case the focus partially rested on integrating public

services that have large volumes of messages. The adoption of the ISD by public service providers was meant "to be the catalyst for other entities. The small ones let's say." (E10.1) Messages with high volumes related to personal income taxes, pensions, child allowances, and the regional invitation to get vaccinated for COVID-19. On the other hand, the architecture of the eBox was set up for integration of eBoxes with private service providers. In this way natural persons could receive invoices from private legal entities (e.g., utility companies) in a single digital environment.

In the My Citizen Profile Case, the following collaboration strategies were followed to deal with the adoption paradox. First, opening up as much personal information contained in (base) registries that could be easily displayed to natural persons to quickly create volume. Second, focus on those services that are useful to a large number of users, such as showing information from local governments with respect to civil registration certificates. Third, prioritizing large public service providers that could more easily integrate a high volume of public services. Fourth, building blocks and functionalities were chosen to reflect the portal functionalities that had not yet been developed. These included a notification functionality, a status update functionality for running procedures and a feedback functionality regarding wrong or incomplete personal data. Fifth, prioritizing authentication for users. This way, users could use the My Citizen Profile to authenticate their identity (and choose the appropriate role) to get single sign-on access to public services. "What we often see is that connecting to [the central access management building block] is a trigger to integrate the header. More and more local governments are starting to work with [it] and you can feel that this causes the [integration to the] header to pick up speed."

In the Digital Invoicing Case the issue of the adoption paradox appeared to play less of a role. Clear rules were established by both the federal and regional governments that gradually mandated the use of the digital public service on both sides of the service chain. Collaboration strategies still had to be devised to motivate public legal entities. To this end, a collaboration strategy of testing the digital invoicing service was used to demonstrate its viability. The testing focused on private and public legal entities with large volumes. A collaboration strategy to further adoption was to implement a flexible infrastructure that allowed suppliers and buyers to send or receive digital invoices without having to rely on each other capabilities.

6. Discussion

6.1. Collaboration strategies for ISD

The results brought forward a large variety of inter-organizational collaboration strategies for ISD directed at many goals. Those collaboration strategies are presented in Table 2. Following the research background, some of these collaboration strategies have been noted, while other collaboration strategies extend the literature. The cases follow the literature that the elements underpinning collaboration strategies can be part of collaboration strategies serving different goals. For example, direct and indirect incentives appear to be important to engage, motivate and create joint action potential alike (Chen & Lee, 2018; van Os, 2011). Indirect incentives can be incorporated through a business case collaboration strategy to make clear the expected benefits (Luna-Reyes et al., 2007). Such a collaboration strategy can provide potential public service providers with a cost-benefit analysis for costs saved or avoided by not developing or managing functionalities themselves. The rest of this section discusses the collaboration strategies related to each of the categories and considers the relationship between collaboration strategies and ISDs.

6.1.1. Engagement collaboration strategies

Collaboration strategies related to coming to principled engagement appeared the least contested in the cases. The cases acknowledge the

importance of lead organizations having formal coordinator roles (Gil-Garcia & Sayogo, 2016; Klievink et al., 2016). Once public service providers had shown a willingness to collaborate, consensus on the division of roles and responsibilities and the position of the lead organizations within the network was relatively quickly reached or accepted, even if the diversity of the collaborating public legal entities was rather high (in the eBox and My Citizen Profile Cases) (Sayogo et al., 2016). The relatively effortless clarity on roles and responsibilities might rather be because of the central position of these organizations within the public administrations. Public service providers joining the collaboration at a later stage also did not appear to pull into question earlier agreements on the division of roles and responsibilities (in line with Gil-Garcia et al., 2019; Lönn & Uppström, 2016). Coming to a shared goal of the ISD and a clearly specified scope was more difficult than agreeing on the distribution of roles. Lead organizations organized interactions with participants collectively and bilaterally to stimulate mutual adjustment (van Os, 2011). Some public service providers in the My Citizen Profile Case who did not see their priorities reflected in the scope took on a lukewarm approach to collaboration or even dropped out.

6.1.2. Motivation collaboration strategies

Collaboration strategies in the cases associated with bringing about motivation seem to be less related to trust creation than is indicated by the literature (Chen, 2008). The eBox law was meant to increase trust between administrations (Luna-Reyes et al., 2007; Yang & Maxwell, 2011). Given that ISDs increase the dependencies of public service providers on public service intermediaries in the chain, trust between collaborators might have to go alongside with formal agreements that set service levels. A department head specified that "trust is sufficient for me at my level, but is not sufficient at the operational level. [...] There have to be agreements, because I also have a customer internally. And if my people can't get to data for three days, then that's a problem for our operational functioning, so there must be service level agreements." (M16.1). Formal agreements were adopted to this extent in the eBox and My Citizen Profile Cases. As a demonstration of each collaboration's viability to deliver public services (Gil-Garcia et al., 2005), the three cases also used early adopters to demonstrate the capabilities necessary to allow ISD and generate trust.

6.1.3. Joint action collaboration strategies

Collaboration strategies related to joint action were the most prominent in the interviews. In line with previous literature (Gil-Garcia & Sayogo, 2016; Pardo et al., 2010), the sharing of information and knowledge was seen as an important strategy to come to joint action. Following Klievink et al. (2016) a first collaboration strategy in this regard related to providing varied documentation and technical support by the lead organizations. A second collaboration strategy, especially acknowledged by respondents in the Digital Invoicing Case, was the transfer of knowledge within and outside of the formal collaboration structure (following Janssen et al., 2012). This is likely more in this case than the other two due to its more limited scope. Respondents indicated that informal collaboration happened when two public service providers relied on the same ERP technology (provider).

Deliberation to come to the voluntary adoption of standards, another collaboration identified by Klievink et al. (2016) was not found in the cases as a collaboration strategy. This could be because of the public-private context of the ISD in which those deliberations took place. In the Digital Invoicing Case, invoice standards were set by the European Directive. In the My Citizen Profile Case, public service providers did partake in processes to develop standards (e.g., related to sending notifications). But those did not appear to be part of the collaboration strategies. Many of the collaboration strategies related to joint action are also outside of those found in the review of the literature. Several collaboration strategies are linked to the characteristics of ISD.

6.1.4. The relationship between collaboration strategies and ISDs

The results demonstrate how collaboration strategies incorporate the characteristics of ISDs and their digital public service chains to enhance collaboration. Collaboration leads to back-office systems of various public service providers being integrated into service chains to achieve ISD. Such an architecture creates dependencies among public service providers. These dependencies include resource dependence, shared user relationships, and shared reputations (Malone & Crowston, 1994), potentially leading public service providers to resist participating in or continuing collaborative efforts. Employing the architecture design strategies put forward by Van Veenstra et al. (2009) shows how the cases incorporated the characteristics of the service chains into their collaboration strategies.

The approach in the eBox Case can be characterized as a broad-casting strategy. A federated system of databases was set up with a frontend eBox application interfacing with several back-end databases based on common standards. The case used the architecture of the public service as a strategy to preserve the autonomy of public service providers through the ISD architecture to generate engagement. This approach was thought to lower the barriers for participation. Public service providers could thus organize (or keep organizing) their own message database. The case also employed the use of formal collaboration agreements at the management level between the lead federal organizations and regional lead organizations to ensure the continuity of message delivery.

The Digital Invoicing Case follows a merger strategy. The lead organization at the federal level owns the main building block through which all incoming and outgoing invoice documents pass through. Invoice documents are standardized through the European Invoicing Directive and information systems are tightly coupled. The federal and regional public legal entities relied on several motivation collaboration strategies and joint action collaboration strategies to manage resource dependency among actors and establish a good reputation of the ISD. These collaboration strategies relate to early adopters and service providers with large volumes to demonstrate the robustness of the IDS and clear communication, transparency, and long-term continuity and financing.

The architectural strategy in the My Citizen Profile Case can be described as an orchestration strategy. Public service providers align their processes with the relevant building blocks of the ecosystem owned and orchestrated by the regional lead organization. A collaboration strategy was chosen in which the public service providers preserve their autonomy. For example, a notification that is intended as a reminder for an appointment only requires a standardized notification at the end of the process. Public service providers can choose between a push and pull model and can also use additional communication channels outside of the ecosystem.

The collaboration process also has to be able to take into account the technical capabilities of the public legal entities. The three cases relied on flexible architectures to allow actions to be taken even though particular public service providers only had a low organizational readiness (Van Veenstra et al., 2009). This enabled lead organizations and public service providers to expand functionalities earlier or tailor solutions to specific needs (e.g., related to legacies or specific requirements), preserving motivation among public service providers and stimulating joint action.

The results indicate the difficulty in creating and maintaining collaboration. The cases highlight how in the absence of a directive regulatory framework (for two of the three cases), prevents finding shared motivation in the early phases. The Digital Invoicing Case shows that pending regulation can be a driver to start collaboration. However, collaboration strategies were still necessary to establish linked service chains and expand functionalities. The findings also point to the limitations of a collaborative approach outside the context of ex ante obligations that drive collaboration on the one hand and adoption on both sides of the service chain on the other hand. This is demonstrated by

governments using their authority to stimulate the adoption of the eBox (at the federal level) and My Citizen Profile (at the regional level) in respectively the operationalization and expansion phases. In a more directive regulatory context, lead organizations in the Digital Invoicing Case had to be less reliant on formalized relationships and embarked on a more flexible and goal-oriented approach (Gil-Garcia et al., 2007; Gil-Garcia et al., 2019).

There seem to be differences in collaboration strategies over the different phases of ISD. During the initiation and expansion phases, a demand-oriented approach appears to be more successful for shared motivation. In the piloting and operationalization phases, a supply-oriented approach to collaboration could potentially lead to more joint action and quicker implementation. However, this could not be analyzed due to the cross-sectional time horizon.

6.2. Limitations

Our research is not without limitations. First, the single-country approach limits the generalizability. The nature of Belgium's federalism poses limitations to the generalizability of the research. The potential for hierarchical coordination is limited as there is no hierarchy of norms between the federal level and the regional entities. This makes collaboration the default option for cases that have an intergovernmental dimension and limits the potential for contrasting cases (Yin, 2018). The research also focused on collaboration strategies, which largely excluded an analysis of the public service context that can actively shape collaboration efforts (Luna-Reyes et al., 2007). Other cases in different contexts might find additional collaboration strategies.

Second, following the choices made in the case selection, the cases we examined presented only a subset of ISDs. For example, the ISDs in this paper were directed at natural persons and private legal entities as users. Many existing ISDs are also between public legal entities, such as data exchange services, or involve co-creation and co-production with natural persons and private legal entities. Other cases might thus reveal different collaboration strategies. As such, we recommend generalizing the findings in further research. At the same time, the cases provide a comprehensive overview of collaboration strategies, with many shared between cases. This leads us to expect that the identified collaboration strategies might be generalizable.

Third, the research is limited regarding the effectiveness of individual and combined collaboration strategies. Further research can look into the effectiveness of individual collaborative strategies in single cases or through a comparative approach of cases. We also recommend examining the impact of ICT-mediated collaboration strategies vs. those that do not.

A fourth limitation relates to the selection of the interviewees. While we adopted a sampling strategy that aimed to be representative to the public legal entities involved in the collaboration, multiple persons in different roles could be involved for a single public legal entity. Some of these persons had already left the public legal entity at the time of the data gathering. Future research could include the roles through which different persons collaborate on behalf a public legal entity, their dynamics and the impact on the collaboration.

Fifth, the cross-sectional time horizon did not allow us to fully discern differences between development phases. It is easy to imagine that initiation might need different collaboration strategies than operationalization and expansion. Research with a longitudinal time horizon (the temporal dimension of collaboration in Chun et al., 2012) could untangle changing types of collaboration strategies across phases, and reveal which collaboration strategies are more suited in the short and long term.

6.3. Contribution to theory

This paper presents the results from three in-depth case studies. These results provide the first comprehensive overview of concrete collaboration strategies in interorganizational collaboration networks to develop and advance ISD, which was lacking in the literature. The results indicate that the identified collaboration strategies might be generalizable, but additional empirical research should be performed. Together with identifying a key issue for advancing ISDs, i.e. the issue of the adoption paradox, the research presented in this paper moves forward the thinking on collaboration strategies in ISD and allows to come to a better understanding of how collaboration unfolds to advance ISDs.

6.4. Implications for practice

ISD remains one of the central digital government challenges. The research points towards collaboration strategies that might prove valuable for practitioners. In line with Gil-Garcia and Sayogo (2016) formally appointed service chain coordinators as lead organizations, either one for the entire service or one per level of government, might prove beneficial to devise and put collaboration strategies into action. An approach centering around lead organizations and bilateral coordination could be an alternative to formal intergovernmental collaboration structures. Understanding and utilizing collaboration strategies adapted to deal with issues such as the adoption paradox, as well as different capacities, capabilities and perspectives to users and public service providers might enable practitioners to further develop, operationalize and expand ISDs.

7. Conclusions

ISD is high on the agenda of governments, but no comprehensive overview of collaboration strategies exists to advance ISD. This article contributes to the e-government literature by identifying collaboration strategies to advance ISD in interorganizational collaboration networks. Three ISD cases in the region of Flanders, Belgium were examined to this extent. We empirically identified 33 collaboration strategies for ISD categorized into engagement, motivation, and joint action collaboration strategies. The contribution is twofold. First, the analysis gives insight into what concrete collaboration strategies are used in ISD and provides insight into how they are formed and combined to address certain issues. Collaboration strategies depend on the situation, as the characteristics of the service chains can impact public service providers' willingness to collaborate and rely on ISD for their service delivery.

In our cases principal collaboration strategies were centered around lead organizations that combined collaboration strategies attuned to each case and often took on intermediary roles in the digital service chains. Lead organizations had to identify the mix of collaboration strategies that advances ISD in the short and long term, as collaboration strategies can be conflicting and public service providers can become less motivated to participate or drop out entirely. This seems especially so in a federal setting with a horizontal distribution of power and without strong intergovernmental collaboration structures.

An important issue was overcoming the 'adoption paradox', which required a combination of collaboration strategies. This issue concerns the impact the adoption of a public service by users has on the adoption by public service providers and vice-versa. Not overcoming this paradox can be seen as a bottleneck for widespread use. Collaboration strategies in the cases included flexible infrastructures that decoupled the capabilities of users and public service providers so one actor can make use of the ISD independent of the technical capabilities of others, and a focus on the integration of public services with high volumes to stimulate public legal entities to collaborate and integrate their services to the ISD. The cases show that collaboration strategies can bring about ISD and provide policy-makers strategies to advance ISD.

As a second contribution, the collaboration strategies identified in this research offer opportunities for research to further expand the knowledge on inter-organizational collaboration that advances ISD. Together with additional cases and further analysis, this research could be used as a stepping stone towards a theoretical model to understand

how collaboration strategies can be combined, prioritized and changed over time to advance ISD. In building towards such a model, further research can focus on finding additional collaboration strategies, examine the effectiveness of collaboration strategies, look into how these collaboration strategies complement each other, and how changing collaboration strategies as ISDs pass through service design and delivery phases. A better empirical understanding of collaboration strategies can lead to insights for academics and practitioners that might advance the long-held dream of ISD.

CRediT authorship contribution statement

Stijn Wouters: Project administration, Writing - original draft,

Conceptualization, Methodology, Investigation, Formal analysis, Visualization. Marijn Janssen: Conceptualization, Formal analysis, Writing – review & editing, Visualization. Veiko Lember: Conceptualization, Writing – review & editing. Joep Crompvoets: Funding acquisition, Project administration, Supervision, Resources, Conceptualization, Writing – review & editing.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Appendix A. Overview of the interviews

ID ^a	Level	Public legal entity	Function interviewee	Interview shared with
D1	Federal	Digital Transformation office	D1.1 Project manager	
			D1.2 Project manager	
D2	Regional	Department	D2.1 Advisor	
D3	Federal	Federal Public Service	D3.1 Team leader	
			D3.2 Project manager	
D4	Federal	Federal Agency	D4.1 Director-general	
		gy	D4.2 Project manager	
D5	Regional	ICT Support Agency	D5.1 Project manager	
	regional	Tot support ilgency	D5.2 Project manager	
D6	Regional	Agency	D6.1 Analyst	
D7	Regional	Agency	D7.1 Team leader	
<i>3</i> /	regional	rigency	D7.1 Team reader	
D8	Regional	Digital Agency	D8.1 Product owner	
D6 D9				
	Regional	Agency	D9.1 Accountant	
D10	Local	City	D10.1 Accounting manager	
D11	Regional	Agency	D11.1 Accounting manager	
D12	Regional	Agency	D12.1 Accountant	
			D12.2 Accountant	
			D12.3 Accountant	
D13	Regional	Agency	D13.1 Department head	
			D13.2 Advisor	
			D13.3 Accountant	
D14	Local	City	D14.1 Process manager	
			D14.2 Accountant	
			D14.3 Business analyst	
E1	Federal	Digital Transformation office	E1.1 Domain manager	M1
E2	Regional	Digital Agency	E2.1 Product owner	
E3	Regional	Digital Agency	E3.1 Analyst	
	_		E3.2 Legal expert	
E4	Federal	Digital Transformation office	E4.1 Project manager	
E5	Regional	ICT Support Agency	E5.1 Program manager	M5
E6	Regional	Agency	E6.1 Department head	
	Ü	5	E6.2 Project manager	
E7	Federal	Federal Public Service	E7.1 Advisor	
_,			E7.2 Project manager	
E8	Regional	Department	E8.1 Department head	M11
LO	regional	Беригинен	E8.2 Project manager	14111
			E8.3 Project manager	
E9	Regional	Agency	E9.1 Manager	
E10	Regional	ICT Support Agency	E10.1 Business analyst	
E10 E11	Federal		•	
211	Federal	Agency	E11.1 Project leader	
F10	P- 41	Ci-1 Ci+ O	E11.2 Team leader	
E12	Federal	Social Security Office	E12.1 Advisor-general	
			E12.2 Program manager	
E13	Federal	Federal Public Service	E13.1 Strategy analyst	
E14	Regional	Agency	E14.1 Program manager	M18
			E14.2 Program architect	
E15	Regional	Agency	E15.1 Manager	M19
E16	Federal	Digital Transformation office	E16.1 Legal expert	
E17	Regional	Agency	E17.2 Account manager	
E18	Federal	Agency	E18.1 Program manager	
			E18.2 Service owner	
M1	Federal	Digital Transformation office	M1.1 Domain manager	E1
M2	Regional	Digital Agency	M2.1 Legal expert	
М3	Regional	Digital Agency	M3.1 Program manager	
M4	Regional	Digital Agency	M4.1 Program manager	
	0			(continued on next page
				COMMINEA ON NEXT DAGE

(continued on next page)

(continued)

ID ^a	Level	Public legal entity	Function interviewee	Interview shared with
M5	Regional	ICT Support Agency	M5.1 Program manager	E5
M6	Regional	Agency	M6.1 Manager	
M7	Regional	Agency	M7.1 Team leader	
M8	Regional	Department	M8.1 Product owner	
			M8.2 Program manager	
M9	Local	City	M9.1 Coordinator	
M10	Regional	Agency	M10.1 Manager	
			M10.2 Project manager	
M11	Regional	Department	E11.1 Department head	E8
			E11.2 Project manager	
			E11.3 Project manager	
M12	Local	City	M12.1 Team leader	
M13	Regional	Department	M13.1 Data and application manager	
M14	Regional	Agency	M14.1 Project manager	
M15	Local	City	M15.1 Project leader	
M16	Regional	Department	M16.1 Division head	
M17	Regional	Agency	M17.1 Team leader	
M18	Regional	Agency	M18.1 Program manager	E14
			M18.2 Program architect	
M19	Regional	Agency	M19.1 Manager	E15
M20	Regional	Agency	M20.1 Product owner	
			M20.2 Product owner	
			M20.3 Product owner	
			M20.4 Coordinator	
M21	Local	City	M21.1 Department head	
M22	Local	Organization	M22.2 Advisor	
M23	Regional	Agency	M23.1 Coordinator	
			M23.2 Team leader	
M24	Local	Municipality	M24.1 Department head	
M25	Local	Municipality	M25.1 Policy manager	

^a D: Digital Invoicing Case, E: eBox Case, M: My Citizen Profile Case.

Appendix B. Interview protocol

Context

What are current or former initiatives related to the public service?

What are the goals and what is the scope of the public service?

What is the general historical context of the public service?

Why and when did your organization became involved in the governance of the public service? What is the role of your organization?

How was the public service initiated inside your organization?

How does the public service stand in relation to the vision concerning user-centricity in your organization?

Actors

Who are the actors?

What are their roles? How are the actors involved in the service chain? Who are the users?

Organization of the public service

What are the involved building blocks and what are their functionalities?

Which choices did your organization make regarding integration and why were does choices made?

What are the functionalities the public service (still) has to provide for your organization?

Governance aspects

What are the strategic and operational goals related to the public service?

How does your organization look at the legal and privacy aspects related to the public service?

What are the organizational aspects related to the public service?

Who is involved in the coordination of the public service?

What are the most important collaboration structures? Are their bilateral consultations?

Which subjects are discussed where?

How and where are decision taken? Are these binding?

What was the impact of the collaboration during initiation/piloting on the current collaboration?

Was and is their consensus on what the scope of the public service is?

Was and is their consensus on how the collaboration should function?

How did and does the collaboration process unfold?

Did the collaboration have an impact on coming to a consensus on the scope?

Should some subjects receive more attention?

What are the different needs and preferences between actors?

What was and is the common motivation to collaborate? Is this motivation shared by all involved actors?

Was and is their consensus on the actions that should be taken?

Have there been obstacles to implementing those actions? How have they been dealt with?

Are there any additional actions that are needed regarding adoption by public legal entities and users?

How does your organization look at contracts and agreements related to the public service?

How does your organization evaluate the collaboration?

(continued on next page)

(continued)

Context

What data does the public service use?

How does your organization look at the semantic and technical standards related to the public service?

How does your organization look at the financial arrangements related to the public service?

General evaluation and future

Did the collaboration have an impact on the outcome of the public service?

Did the collaboration have an impact on how your organization views user-centricity, generic building blocks and/or ISD?

What are the follow-up steps?

Are there elements we haven't discussed yet but that you want to add?

Appendix C. Frequency of references to collaboration strategies

Collaboration strate	gies		Digital Invoicing	eBox	My Citizen Profile
Related to engagement	Open deliberation process of actors involved in collaboration efforts	(11/ 17) ^a	Yes (1/1)	Yes (2/2)	Yes (8/14)
0.0	Preserving the autonomy of public service providers through the architecture	(5/6)	No (n.m.) ^b	Yes (4/5)	Yes (1/1)
	Business case to focus collaboration	(7/12)	No (1/2)	Yes (4/5)	Yes (2/5)
	Formal agreements at the management level to settle roles and responsibilities	(17/ 30)	No	Yes (10/22)	Yes (7/8)
	Use cases, pilots and proof of concepts	(14/	(n.m.) Yes	Yes	Yes
	Deliberate use of names	16) (5/5)	(5/6) Yes	(5/6) No	(4/4) Yes
	Demand oriented approach to deliberation, with priorities set by priorities by lead	(9/9)	(1/1) No	(n.m.) Yes	(4/4) Yes
	organization(s) Supply oriented approach to deliberation	(14/	(n.m.) Yes	(3/3) No	(6/6) Yes
Related to	Formal agreements at the management level to build trust and commitment	27) (6/6)	(7/16) No	(n.m.) Yes	(7/11) Yes
motivation	Finding common motivation and mutual understanding of each other's context	(10/	(n.m.) Yes	(4/4) Yes	(2/2) Yes
	Stimulate competition among early adopting public service providers	13) (6/9)	(1/2) Yes	(6/7) No	(3/4) Yes
	Focusing on public services with large volumes or important public services	(17/	(5/8) Yes	(n.m.) Yes	(1/1) Yes
	Giving financial incentives	33) (12/	(1/1) Yes	(8/15) No	(8/17) Yes
	Emphasizing lower burden for public service providers	18) (13/	(4/5) Yes	(n.m.) Yes	(8/13) Yes
	Use early adopters to demonstrate capability and build trust	18) (6/7)	(1/1) Yes	(6/9) Yes	(6/8) Yes
	Being transparent about choices that are made	(9/17)	(1/1) Yes	(2/2) Yes	(3/4) Yes
	Making clear long term continuity and finance	(8/10)	(2/5) Yes	(1/1) No	(6/11) No
Related to joint	Dyadic collaboration by lead organizations	(35/	(8/10) Yes	(n.m.) Yes	(n.m.) Yes
action	Separating business and IT collaboration	76) (9/15)	(6/6) Yes	(11/41) Yes	(18/29) Yes
	Re-using already existing services	(4/5)	(5/9) /°	(1/2) In part (2/	(3/4) Yes
	Change structural coordination approach across phases	(17/	No	2) Yes	(2/3) Yes
	Flexible user-oriented architecture (regarding the users)	34) (12/	(n.m.) Yes	(9/17) Yes	(8/17) /
	Core functionalities and stepwise expansion	22) (12/	(7/14) Yes	(5/8) Yes	No
	Flexible architecture, approach and legacy management regarding (adoption by) public	18) (36/	(3/3) Yes	(9/15) Yes	(n.m.) Yes
	service providers Giving financial incentives to (early adopting) public service providers	91) (14/	(6/13) Yes	(16/37) No	(14/41) Yes
	Provide clear roadmap(s) to focus and to anticipate changes	17) (16/	(5/5) Yes	(n.m.) Yes	(9/12) Yes
	Common support (for the integration of the various involved building blocks)	25) (13/	(2/2) Yes	(7/14) Yes	(7/9) Yes
	Taking into account financial and technical capabilities	16) (16/	(3/4) Yes	(6/7) Yes	(4/5) Yes
	Bilateral agreements (with public service providers / intermediaries / lead organizations)	20) (35/ 57)	(4/4) Indirect (10/13)	(4/4) Yes (13/27)	(8/12) Yes (12/17)

(continued on next page)

(continued)

Collaboration strategies		Digital Invoicing	eBox	My Citizen Profile
Stimulating knowledge transfer between involved (or new) public service providers wi		Yes	In part	Yes
and outside of collaboration	21)	(9/19)	(1/1)	(1/1)
Providing varied documentation and technical support	(21/	Yes	Yes	Yes
	36)	(8/17)	(6/11)	(7/8)
Using legal frameworks to come to institutional arrangements / settle legal uncertainty	(9/14)	No	Yes	Yes
		(n.m.)	(8/13)	(1/1)
Ensuring a sustainable relation with IT-providers	(14/ 32)	In part (5/7)	Yes (4/5)	Yes (5/20)

- ^a (Number of interviews with mentions / total number of mentions).
- ^b Not mentioned in the interviews.
- ^c Not a feasable collaboration strategy due to the architecture of the ISD.

References

- Agranoff, R. (1991). Human services integration: Past and present challenges in public administration. Public Administration Review, 51(6), 533–542. https://doi.org/ 10.2307/976604
- Agranoff, R. (2017). Crossing boundaries for intergovernmental management. Washington, D.C.: Georgetown University Press.
- Bharosa, N., van Wijk, R., de Winne, N., & Janssen, M. (2015). Challenging the chain: Governing the automated exchange and processing of business information. Delft: IOS Press. https://doi.org/10.3233/978-1-61499-497-8-i
- Bhattacherjee, A. (2012). Social science research: Principles, methods, and practices. Tampa: Global Text Project.
- Boudreau, C., & Bernier, L. (2017). The implementation of integrated electronic service delivery in Quebec: The conditions of collaboration and lessons. *International Review* of Administrative Sciences, 83(3), 602–620. https://doi.org/10.1177/ 0020852315598215
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology*, 3(2), 77–101. https://doi.org/10.1191/1478088706qp063
- Bryson, J., & George, B. (2020). Strategic management in public administration. In B. G. Peters, & I. Thynne (Eds.), *The Oxford encyclopedia of public administration*. Oxford: Oxford University Press. https://doi.org/10.1093/acrefore/9780190228637.013.1396.
- Bryson, J. M., Crosby, B. C., & Stone, M. M. (2006). The Design and Implementation of Cross-Sector Collaborations: Propositions from the Literature. *Public Adminstration Review*, 66(1), 44–55. https://doi.org/10.1111/j.1540-6210.2006.00665.x
- Chen, B. (2008). Assessing interorganizational networks for public service delivery: A process-perceived effectiveness framework. *Public Performance & Management Review*, 31(3), 348–363. http://www.jstor.org/stable/20447681.
- Chen, Y.-C., Hu, L.-T., Tseng, K.-C., Juang, W.-J., & Chang, C.-K. (2019). Cross-boundary e-government systems: Determinants of performance. *Government Information Quarterly*, 36(3), 449–459. https://doi.org/10.1016/j.giq.2019.02.001
- Chen, Y.-C., & Lee, J. (2018). Collaborative data networks for public service: Governance, management, and performance. *Public Management Review*, 20(5), 672–690. https://doi.org/10.1080/14719037.2017.1305691
- Chun, S. A., Luna-Reyes, L. F., & Sandoval-Almazán, R. (2012). Collaborative e-government. Transforming Government: People, Process and Policy, 6(1), 5–12. https://doi.org/10.1108/17506161211214868
- Dawes, S. S., Cresswell, A. M., & Pardo, T. A. (2009). From "need to know" to "need to share": Tangled problems, information boundaries, and the building of public sector knowledge networks. *Public Administration Review*, 69(3), 392–402. https://doi.org/10.1111/j.1540-6210.2009.01987 2.x
- Dedrick, R. F., & Greenbaum, P. E. (2011). Multilevel confirmatory factor analysis of a scale measuring interagency collaboration of Children's mental health agencies. *Journal of Emotional and Behavioral Disorders*, 19(1), 27–40. https://doi.org/ 10.1177/1063426610365879
- Denscombe, M. (2014). The good research guide: For small-scale social research projects.

 Maidenhead: McGraw-Hill Education.
- Dunleavy, P., Margetts, H., Bastow, S., & Tinkler, J. (2005). New public management is dead – Long live digital-era governance. *Journal of Public Administration Review and Theory*, 16(3), 467–494. https://doi.org/10.1093/jopart/mui057
- Einbinder, S. D., Robertson, P. J., Garcia, A., Vuckovic, G., & Patti, R. J. (2000). Interorganizational collaboration in social service organizations: A study of the prerequisites to success. *Journal of Children and Poverty*, 6(2), 119–140. https://doi. org/10.1080/713675966
- Emerson, K., & Nabatchi, T. (2015). Collaborative governance regimes. Washington, D.C.: Georgetown University Press.
- Fan, B., Liu, R., Huang, K., & Zhu, Y. (2019). Embeddedness in cross-agency collaboration and emergency management capability: Evidence from Shanghai's urban contingency plans. *Government Information Quarterly*, 36(4), Article 101395. https://doi.org/10.1016/j.giq.2019.101395
- Gasco-Hernandez, M., Gil-Garcia, J. R., & Luna-Reyes, L. F. (2022). Unpacking the role of technology, leadership, governance and collaborative capacities in inter-agency collaborations. Government Information Quarterly. https://doi.org/10.1016/j. giq.2022.101710 (in press), 101710.
- Gil-Garcia, J. R. (2012). Towards a smart state? Inter-agency collaboration, information integration, and beyond. *Information Polity: The International Journal of Government &*

- Democracy in the Information Age, 17(3), 269–280. https://doi.org/10.3233/IP-2012-000287
- Gil-Garcia, J. R., Chengalur-Smith, I., & Duchessi, P. (2007). Collaborative e-government: Impediments and benefits of information-sharing projects in the public sector. European Journal of Information Systems, 16(2), 121–133. https://doi.org/10.1057/palgrave.ejis.3000673
- Gil-Garcia, J. R., Guler, A., Pardo, T. A., & Burke, G. B. (2019). Characterizing the importance of clarity of roles and responsibilities in government inter-organizational collaboration and information sharing initiatives. Government Information Quarterly, 36(4), Article 101393. https://doi.org/10.1016/j.giq.2019.101393
 Gil-Garcia, J. R., & Sayogo, D. S. (2016). Government inter-organizational information
- Gil-Garcia, J. R., & Sayogo, D. S. (2016). Government inter-organizational information sharing initiatives: Understanding the main determinants of success. *Government Information Quarterly*, 33(3), 572–582. https://doi.org/10.1016/j.giq.2016.01.006
- Gil-Garcia, J. R., Schneider, C. A., Pardo, T. A., & Cresswell, A. M. (2005).

 Interorganizational information integration in the criminal justice enterprise:

 Preliminary lessons from state and county initiatives. In Proceedings from the 38th annual Hawaii international conference on system sciences (HICSS) (p. 118c).

 Washington, DC: IEEE Computer Society. https://doi.org/10.1109/HICSS.2005.338.
- Goldkuhl, G. (2012). Pragmatism vs interpretivism in qualitative information systems research. European Journal of Information Systems, 21(2), 135–146. https://doi.org/ 10.1057/ejis.2011.54
- Imperial, M. (2005). Using collaboration as a governance strategy: Lessons from six watershed management programs. Administration and Society, 37(3), 281–320. https://doi.org/10.1177/0095399705276111
- Janssen, M., Kamal, M. M., Weerakkody, V., & Joha, A. (2012). Shared services as a collaboration strategy and arrangement in public service networks. In *Proceedings* from the 45th Hawaii international international conference on systems science (HICSS) (pp. 2218–2227). Washington, DC: IEEE Computer Society. https://doi.org/ 10.1109/HICSS.2012.527.
- Janssen, M., & Klievink, B. (2009). The role of intermediaries in multi-channel service delivery strategies. *International Journal of Electronic Government Research*, 5(3), 36–46. https://doi.org/10.4018/jegr.2009070103
- Johnson, G., Scholes, K., & Whittington, R. (2008). Exploring corporate strategy. Harlow: Prentice Hall.
- Juell-Skielse, G., Lönn, C.-M., & Päivärinta, T. (2017). Modes of collaboration and expected benefits of inter-organizational E-government initiatives: A multi-case study. Government Information Quarterly, 34(4), 578–590. https://doi.org/10.1016/j. giq.2017.10.008s
- Kettl, D. F. (2006). The transformation of governance: Public administration for twenty-first century America. Baltimore: Johns Hopkins University Press.
- Klievink, B., Bharosa, N., & Tan, Y.-H. (2016). The collaborative realization of public values and business goals: Governance and infrastructure of public-private information platforms. Government Information Quarterly, 33(1), 67–79. https://doi.org/10.1016/j.giq.2015.12.002
- Klievink, B., & Janssen, M. (2008). Improving government service delivery with private sector intermediaries. *European Journal of ePractice*, 5(October), 1–9.
- Lang, A., & Brüesch, C. (2020). Collaborative governance in program implementation: The development of E-relocation notification in the Swiss Canton of Zurich. *International Journal of Public Administration*, 43(12), 1083–1095. https://doi.org/ 10.1080/01900692.2019.1665681
- Lindgren, I., & Jansson, G. (2013). Electronic services in the public sector: A conceptual framework. Government Information Quarterly, 30(2), 163–172. https://doi.org/ 10.1016/j.giq.2012.10.005
- Lönn, C., & Uppström, E. (2016). Understanding public sector collaboration through boundary object theory: A case study of an E-government initiative in Sweden. In Proceedings from the 49th Hawaii international conference on system sciences (HICSS) (pp. 2687–2696). Washington, DC: IEEE Computer Society. https://doi.org/ 10.1109/HICSS.2016.337.
- Luna-Reyes, L. F., Gil-Garcia, J. R., & Cruz, C. B. (2007). Collaborative digital government in Mexico: Some lessons from federal web-based interorganizational information integration initiatives. *Government Information Quarterly*, 24(4), 808–826. https://doi.org/10.1016/j.giq.2007.04.003
- Lundin, M. (2007). When does cooperation improve public policy implementation? Policy Studies Journal, 35(4), 629–652. https://doi.org/10.1111/j.1541-0072.2007.00240.x

- Malone, T. W., & Crowston, K. (1994). The interdisciplinary study of coordination. ACM Computing Surveys, 26(1), 87–119. https://doi.org/10.1145/174666.174668
- Miles, M. B., & Huberman, A. M. (1994). Qualitative data analysis: An expanded sourcebook. Thousand Oaks: Sage.
- van Os, G. (2011). The challenge of coordination: Coordinating integrated electronic service delivery in Denmark and the Netherlands. *Information Polity*, 16(1), 51–61.
- Pardo, T. A., Gil-Garcia, J. R., & Luna-Reyes, L. F. (2010). Collaborative governance and cross-boundary information sharing: Envisioning a networked and IT-enabled public administration. In R. O'Leary, D. M. Van Slyke, & S. Kim (Eds.), The future of public administration around the world: The Minnowbrook perspective (pp. 129–139). Washington, D.C.: Georgetown University Press. http://www.jstor.org/stable/j.ctt 214cr 21
- Picazo-Vela, S., Gutiérrez-Martínez, I., Duhamel, F., Luna, D. E., & Luna-Reyes, L. F. (2018). Value of inter-organizational collaboration in digital government projects. *Public Management Review*, 20(5), 691–708. https://doi.org/10.1080/ 14719037.2017.1305702
- Provan, K. G., & Milward, H. B. (2001). Do networks really work? A framework for evaluating public-sector organizational networks. *Public Administration Review*, 61 (4), 414–423. https://doi.org/10.5465/APBPP.1999.27628083
- Puron-Cid, G., Luna, D. E., Picazo-Vela, S., Gil-Garcia, J. R., Sandoval-Almazan, R., & Luna-Reyes, L. F. (2022). Improving the assessment of digital services in government websites: Evidence from the Mexican state government portals ranking. Government Information Quarterly, 39(1), Article 101589. https://doi.org/10.1016/j. giq.2021.101589
- Ritchie, J., Lewis, J., McNaughton Nicholls, C., & Ormston, R. (2013). Qualitative research practice: A guide for social science students and researchers. Thousand Oaks: Sage.
- Sayogo, D. S., Gil-Garcia, J. R., & Cronemberger, F. (2016). Determinants of clarity of roles and re-sponsibilities in interagency information integration and sharing (IIS). In H. J. Scholl, O. Glassey, M. Janssen, B. Klievink, I. Lindgren, P. Parycek, ... D. Sá Soares (Eds.), Vol. 9820. EGOV 2016, LNCS (pp. 126–134). Cham: Springer.
- Scholl, H. J., & Klischewski, R. (2007). E-government integration and interoperability: Framing the research agenda. *International Journal of Public Administration*, 30(8), 889–920. https://doi.org/10.1080/01900690701402668
- Swanborn, P. (2010). Case study research: What, why and how? Thousand Oaks: Sage. https://doi.org/10.4135/9781526485168
- Van Donge, W., Bharosa, N., & Janssen, M. (2022). Data-driven government: Cross-case comparison of data stewardship in data ecosystems. Government Information Quarterly, 39(2), Article 101642. https://doi.org/10.1016/j.giq.2021.101642
- Van Veenstra, A. F., & Janssen, M. (2011). Architectural principles for orchestration of cross-organizational service delivery: Case studies from the Netherlands. In S. Assar, I. Boughzala, & I. Boydens (Eds.), Practical studies in E-government: Best practices from around the world (pp. 167–185). New York Dordrecht Heidelberg London: Springer Science+Business Media. https://doi.org/10.1007/978-1-4419-7533-1_10.
- Van Veenstra, A. F., Janssen, M., & Klievink, B. (2009). Strategies for orchestrating and managing supply chains in public service networks. *Electronic Journal of e-Government*, 7(4), 425–432.
- Vangen, S., & Huxham, C. (2003). Nurturing collaborative relations: Building trust in interorganizational collaboration. *The Journal of Applied Behavioral Science*, 39, 5–31. https://doi.org/10.1177/0021886303039001001

- Waldfogel, J. (1997). The new wave of service integration. Social Service Review, 71(3), 463–484.
- Walsham, G. (1995). Interpretive case studies in IS research: Nature and method. European Journal of Information Systems, 4(2), 74–81. https://doi.org/10.1057/aiii 1005.0
- Wimmer, M. (2002). Integrated service modelling for online one-stop government. Electronic Markets, 12(3), 149–156.
- Wouters, S., Janssen, M., & Crompvoets, J. (2020). Governance challenges of interorganizational digital public services provisioning: A case study on digital invoicing services in BelgiumG. Viale Pereira, M. Janssen, H. Lee, I. Lindgren, M. P. Rodríguez Bolívar, H. J. Scholl, & A. Zuiderwijk (Eds.). In *Electronic Government: EGOV 2020, LNCS, vol. 12219* (pp. 223–235). Cham: Springer.
- Yang, T.-M., & Maxwell, T. A. (2011). Information-sharing in public organizations: A literature review of interpersonal, intra-organizational and inter-organizational success factors. Government Information Quarterly, 28(2), 164–175. https://doi.org/ 10.1016/j.giq.2010.06.008
- Yildiz, M. (2007). E-government research: Reviewing the literature, limitations, and ways forward. Government Information Quarterly, 24(3), 646–665. https://doi.org/ 10.1016/j.giq.2007.01.002
- Yin, R. K. (2018). Case study research and applications. Thousand Oaks: Sage.

Stijn Wouters has a PhD in Social Sciences and is affiliated with the KU Leuven Public Governance Institute. His research interests are related to the governance of e-government and digital public services in multi-governance settings. He works for the Social and Economic Council of Flanders (SERV) on topics including digitalization, the governance of the public administration, administrative simplification, and better regulation.

Marijn Janssen is a full Professor in ICT & Governance at the Information and Communication Technology research group of the Technology, Policy and Management Faculty of Delft University of Technology. He is Co-Editor-in-Chief of Government Information Quarterly, conference chair of IFIP EGOV series and is chairing mini-tracks at e-government and information systems conferences.

Veiko Lember is Senior Research Fellow at Tallinn University of Technology's Ragnar Nurkse Department of Innovation and Governance and Visiting Professor at the KU Leuven Public Governance Institute (Belgium). His main research interests are in public sector innovation, digital governance and innovation policy. Veiko is a Member of Steering Committee of the European Group for Public Administration (EGPA), where he also cochairs its permanent study group on Public Administration, Technology and Innovation (PATI).

Joep Crompvoets is professor and research manager at the KU Leuven Public Governance Institute holding the chair on 'Information Management in the Public Sector'. His research refers to e-governance, public sector innovation, digital transformation in the public sector, GIS and (spatial) data infrastructures. He is secretary-general of EuroSDR - a European spatial data research network linking national mapping agencies with research institutes/universities for the purpose of applied research in the domain of geospatial information management.